Author's response to reviews

Title: Sugar-added beverages consumption among kindergarten children of Crete: effects on nutritional status and risk of obesity

Authors:

Manolis Linardakis (linman@med.uoc.gr)
Katerina Sarri (katsarri@med.uoc.gr)
Maria-Styliani Pateraki (melin Pat@yahoo.com)
Manolis Sbokos (manosmpok@yahoo.gr)
Anthony Kafatos (kafatos@med.uoc.gr)

Version: 5 Date: 17 July 2008

Author's response to reviews: see over
Response to reviewers
The authors would like to thank the reviewers for their comments.

Reviewer: F. Bellisle
Minor revisions
1. Page 7, line 9: It has been corrected.
2. Page 7, lines 10-12: The reference group suggested by the reviewer was the one used by the authors in the first place (please see response number 2 in the file ‘Response to reviewers’ sent on the 28th of March 2008). However, by mistake, the wording was not changed in the text. This has now been corrected.
3. Page 7 line 18: It has been corrected.
4. Page 7 line 21: It has been corrected.
5. Page 9: A reference has been added.
6. Table 1: Notes 1 and 2 refer to the statistical tests used. However, no significant statistical differences were found with the exception of note 3, and that is the reason why note 3 is mentioned separately.
7. The differences in the format of the results (Mean and SD versus mean and SE) have to do with the statistical test used (Analysis of covariance estimates marginal means with standard errors).
8. Table 2: N is now given in all groups within brackets.